Author's response to reviews

Title: Effects of interpretive front-of-pack nutrition labels on food purchases: protocol for the Starlight randomised controlled trial.

Authors:

Ekaterina Volkova (k.volkova@auckland.ac.nz)
Bruce Neal (bneal@georgeinstitute.org.au)
Mike Rayner (mike.rayner@dph.ox.ac.uk)
Boyd A Swinburn (boyd.swinburn@auckland.ac.nz)
Helen Eyles (h.eyles@auckland.ac.nz)
Yannan Jiang (y.jiang@auckland.ac.nz)
Jo Michie (j.michie@auckland.ac.nz)
Cliona Ni Mhurchu (c.nimhurchu@auckland.ac.nz)

Version: 2
Date: 12 September 2014

Author's response to reviews:

Dear Editorial Office for BMC Public Health,

Thank you for the comments provided on the article entitled “Effects of interpretive front-of-pack nutrition labels on food purchases: protocol for the Starlight randomised controlled trial”. Please find revised copy of the manuscript enclosed with this submission.

The following changes have been made in response to the comments received on the 28th August 2014:

1) The abstract was amended to follow the required structure. The abstract does not contain abbreviations and is below 350 words.

2) Trial registration date was added to the abstract

3) Additional supporting documents were removed from the submission

Thank you and we look forward to hearing back from you with the decision regarding our article.

Kind regards,

Ekaterina Volkova PhD
National Institute for Health Innovation;
Level 4, School of Population Health; The University of Auckland (Tamaki Campus)
261 Morrin Road; Glen Innes; Auckland 1142
Phone: +64 9 923 4742
Email: k.volkova@nihi.auckland.ac.nz